Author's response to reviews

Title: Effects of different flow patterns and end-inspiratory pause on oxygenation and ventilation in newborn piglets: an experimental study.

Authors:

Carlos Ferrando (cafeoranestesia@gmail.com)
Marisa García (marisagarciap@gmail.com)
Andrea Gutierrez (andreguti@hotmail.com)
Jose A Carbonell (joseacarbonell19@hotmail.com)
Marina Soro (soromarina@gmail.com)
Gerardo Aguilar (gerardo.aguilar@uv.es)
Javier Belda (fco.javier.belda@uv.es)

Version: 6 Date: 19 September 2014

Author's response to reviews:

Response to the editorial requirements:
A complete checklist of the ARRIVE guidelines has been included in the manuscript.